Author’s response to reviews

Title: Effect of integrating maternal health services and family planning services on postpartum family planning behavior in Ethiopia: Results from a longitudinal survey

Authors:

Linnea Zimmerman (linnea.zimmerman@jhu.edu)
Yuanyuan Yi (yyi7@jhu.edu)
Mahari Yihdego (mahariyihdego@gmail.com)
Solomon Abrha (abrsole2006@gmail.com)
Solomon Shiferaw (solomon.shiferaw@gmail.com)
Assefa Seme (assefa.seme@gmail.com)
Saifuddin Ahmed (sahmed3@jhu.edu)

Version: 3 Date: 25 Sep 2019

Author’s response to reviews:

Thank you for updating the Authors’ contributions section with author initials. We note that there are two authors with the same initials, SA. Please distinguish them either by using numbers - by starting with the author higher up the author list – or adding additional letters. We have distinguished the two authors by including their author number after their initials. Thank you.